Author's response to reviews

Title: Bridging the Gap: partnerships and quality improvement innovations to address refugee maternal and child health inequalities

Authors:

Jane Yelland (jane.yelland@mcri.edu.au)
Elisha Riggs (elisha.riggs@mcri.edu.au)
Josef Szwarc (szwarcj@foundationhouse.org.au)
Sue Casey (caseys@foundationhouse.org.au)
Wendy Dawson (wendy.dawson@mcri.edu.au)
Dannielle Vanpraag (dannielle.vanpraag@mcri.edu.au)
Christine East (christine.east@monash.edu)
Euan Wallace (euan.wallace@monash.edu)
Glyn Teale (Glyn.Teale@wh.org.au)
Bernadette Harrison (bernadette.harrison@cgd.vic.gov.au)
Pauline Petschel (Pauline.Petschel@wyndham.vic.gov.au)
John Furler (j.furler@unimelb.edu.au)
Sharon Goldfeld (sharon.goldfeld@mcri.edu.au)
Fiona Mensah (fiona.mensah@mcri.edu.au)
Mary Anne Biro (maryanne.biro@monash.edu)
Suzanne Willey (suzanne.willey@monash.edu)
I-Hao Cheng (i.cheng@semml.com.au)
Rhonda Small (i.cheng@semml.com.au)
Stephanie Brown (stephanie.brown@mcri.edu.au)

Version: 2 Date: 27 March 2015

Author's response to reviews:

Thank you for your email. The manuscript now includes the title page and the competing interests statement. Additional files contain the NHMRC funding approval letter and the Royal Children's Health Ethics approval.

Kind regards, Jane Yelland